Author's response to reviews

Title: Effects of lung protective mechanical ventilation associated with permissive respiratory acidosis on regional extra-pulmonary blood flow in experimental ARDS

Authors:

Rudolf Hering (Rudolf.Hering@ukb.uni-bonn.de)

Stefan Kreyer (Stefan.Kreyer@ukb.uni-bonn.de)

Christian Putensen (Christian.Putensen@ukb.uni-bonn.de)

Version: 2 Date: 16 Oct 2017

Author's response to reviews:

1. We renamed the heading "Material and Methods" to "Methods" as requested.

2. We adapted the statement "The datasets generated and analysed during the current study are available from the corresponding author on reasonable request".

3. We added a section "Additional files" as requested.

4. We removed any tracked changes or highlighting from the manuscript.